Author’s response to reviews

Title: Angiolymphoid hyperplasia with eosinophilia of the hand: treat with a reverse digital island flap and artificial skin: a case report

Authors:

Naohiro Oka (n-oka@med.kindai.ac.jp)
Shunji Nishimura (shunnisi@med.kindai.ac.jp)
Hiroki Tanaka (hiroki@med.kindai.ac.jp)
Kazuhiko Hashimoto (hazzhiko@med.kindai.ac.jp)
Ryosuke Kakinoki (rkakinoki@med.kindai.ac.jp)
Masao Akagi (makagi@med.kindai.ac.jp)

Version: 2 Date: 16 Feb 2019

Author’s response to reviews:

Dr. Itzhak Brook
Transmitting editor
Journal of Medical Case Reports

Re: JMCR-D-18-00561

Dear Dr. Itzhak Brook

We thank the referee for fruitful suggestions, especially for suggesting the better terms and sentences.

We have revised the manuscript "Angiolymphoid hyperplasia with eosinophilia of the hand: treat with a reverse digital island flap and artificial skin: a case report" (JMCR-D-18-00561) on the basis of the Referee’s comments.

We look forward to a publication of our manuscript in Journal of Medical Case Reports.

Our responses to the referee’s comments are as follows:

We have attached the revised manuscript.
1. Give detailed physical, and neurological examinations results on admission.

Response: We added her physical, and neurological examinations results on page 4. Changes are in red.

2. The manuscript requires extensive editing in style and grammar.

Response: We sincerely thank you for your valuable comment. The manuscript has been carefully reviewed by an experienced editor whose first language is English. We sent the certification of English proofreading.

The original comments of the Referee are as follows.

Reviewer reports:

Thanks for the revisions.

Give detailed physical, and neurological examinations results on admission.

The manuscript requires extensive editing in style and grammar.

We hope that you will consider our new version satisfactory. We believe that we have been successful in providing the corrections and statements to increase the quality, novelty, and general interest of our manuscript. We look forward to your response.

Sincerely, Naohiro Oka.

Department of Orthopedic Surgery, Kindai University, Japan.
377-2 Ohno-Higashi, Osaka-Sayama, Osaka 589-8511, Japan
Tel: +81-72-366-0221
Fax: +81-72-366-0245
e-mail : n-oka@med.kindai.ac.jp